\documentclass
  [
    math-font    = STIX Two Math, agreed,
    CJKmain-font = { {Songti SC}[AutoFakeBold = 2.5, AutoFakeSlant] },
    CJKsans-font = { {STHeiti}[AutoFakeBold = 2] }
  ] {hduthesis}
\tikzset{ > = stealth }
\usetikzlibrary{positioning, shapes.geometric}

\hduset
  {
    title      = HDU å­¦å£«å­¦ä½è®ºæ–‡ \hologo{LaTeX} æ¨¡æ¿ç¤ºä¾‹æ–‡æ¡£/
                 æœ¬ç§‘æ¯•ä¸šè®¾è®¡,
    department = ç†å­¦é™¢,
    major      = ç‰©ç†å­¦,
    class      = è‹±æ‰ç­,
    stdntid    = C668668E,
    author     = ç”³æ™ºèƒ½,
    supervisor = æ•™æŽˆ:è‘‰èŠ·æ™´,
    % bibsource  = xampl
  }

\begin{document}

\maketitle
\commitment [ example-image-a/2024-05-31 ]

\begin{abstract}[cn]
  \keywords{æ­å·žç”µå­ç§‘æŠ€å¤§å­¦, æ¯•ä¸šè®ºæ–‡, \hologo{LaTeX3}, }
\end{abstract}

\begin{abstract}[en]
  \keywords{HDU, thesis, \hologo{LaTeX3}, }
\end{abstract}

\tableofcontents

\chapter{å¼•è¨€}
1

\cite{whole-journal}

\zhlipsum[2]
\cite{inbook-crossref,whole-set,booklet-full,incollection-crossref,whole-collection,manual-full,mastersthesis-full}

\chapter{å…¬å¼ä¸Žæ’å›¾æµ‹è¯•}

\begin{equation}
  (i\gamma^\mu\partial_\mu - m)\psi = 0
  \label{2-1}
\end{equation}
å…¶ä¸­ \eqref{2-1} ä¸ºDirac æ–¹ç¨‹.
\[
  S_x = \frac12\sigma_1 = \frac12
  \begin{pmatrix}
    0 & 1\\
    1 & 0
  \end{pmatrix}, \quad
  S_y = \frac12\sigma_1 = \frac12
  \begin{pmatrix}
    0 & -i\\
    i & 0
  \end{pmatrix}, \quad
  S_z = \frac12\sigma_1 = \frac12
  \begin{pmatrix}
    1 & 0\\
    0 & -1
  \end{pmatrix}
\]
ä¸Šå¼ä¸º Pauli çŸ©é˜µ.

\printbibliography

\appendix
\chapter*{é™„å½•}
\addcontentsline{toc}{chapter}{é™„å½•}

\end{document}
